Author's response to reviews

Title: Clinicians' caseload management behaviours as explanatory factors in patients' length of time on caseloads: a predictive multilevel study in paediatric community occupational therapy

Authors:

Niina Kolehmainen (n.kolehmainen@abdn.ac.uk)
Graeme MacLennan (g.macclennan@abdn.ac.uk)
Jillian J Francis (j.francis@abdn.ac.uk)
Edward AS Duncan (edward.duncan@stir.ac.uk)

Version: 5 Date: 18 August 2010

Author's response to reviews: see over
Dear BMC Series Editorial Production Team,

Thank you for your assistance in preparing our manuscript for publications. We have made the corrections suggested in your latest email. Please see details below.

Please do not hesitate to contact me if you would require us to make any further changes.

Kind regards,

Niina Kolehmainen

**Major revisions**

Reference 10: please provide a functioning web link to the website.

_Apparently the organization hosting this link has recently re-designed their web page and they no longer provide access to this reference on their web page. I have revised the reference in the list to reflect this change._

Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.

_My apologies for the shading that remained. For some reason this did not show on my computer; all shading should now have been removed._

**Minor revisions**

Tables- please remove the visible vertical lines from your tables

_This has been done._

Figures: It is important for the final layout of the manuscript that the figures are in the correct orientation. Our online figure guide contains full details for preparing files for submission and can be viewed here: [http://www.biomedcentral.com/info/ifora/figures](http://www.biomedcentral.com/info/ifora/figures)

_I was slightly unsure what was meant by this; I have carefully re-read the guidelines and although I could not find anything stating that the figures should be in portrait I wondered if this was what was meant. So I have changed the figure into portrait. I hope this will be helpful._

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

_We have checked this and are happy with the manuscript._